Author's response to reviews

Title: Eradication of intractable malignant ascites by abdominocentesis, reinfusion of concentrated ascites, and adoptive immunotherapy with dendritic cells and activated killer cells in a recurrent lung cancer patient.

Authors:

Hideki Kimura (hkimura@chiba-cc.jp)
Toshihiko Iizasa (tiizasa@chiba-cc.jp)
Aki Ishikawa (aishikawa@chiba-cc.jp)
Mitsuru Yoshino (myoshino@chiba-cc.jp)
Masato Shingyouji (mshingyoji@chiba-cc.jp)
Masaki Kimura (mkimura@chiba-cc.jp)
Tetushi Hirata (thirata@chiba-cc.jp)
Akiko Odaka (aodaka@chiba-cc.jp)
Keiko Matsubayasi (kmatsuba@chiba-cc.jp)

Version: 2 Date: 2 December 2007

Author's response to reviews:

Hideki Kimura, M.D., Ph.D.
Division of Thoracic Diseases,
Chiba Cancer Center,
Nitona-cho 666-2, Chuo-ku, Chiba 260-8717, Japan

Des, 2, 2007
To: Editorial Board, BMC Cancer

Dear Editorial Board

This is an original report entitled "Eradication of intractable malignant ascites by abdominocentesis, reinfusion of concentrated ascites, and adoptive immunotherapy with dendritic cells and activated killer cells in a recurrent lung cancer patient." by Hideki Kimura, Toshihiko Iizasa, Aki Ishikawa et al. which we are submitting for publication in Cancer. We believe that this report will be of interest to fellow researchers and clinicians in the field of cancer therapy. I have been chosen to represent all of the authors and accept responsibility for the release of this material. I certify that the article is original, is not under consideration by any other journals, has not been previously published, and that the data in the manuscript have been reviewed by all of the authors, who agree with the analysis and the conclusions reached therein.

I am very glad that we can apply our manuscript to the leading journal in this field. I hope you will find the manuscript acceptable for publication in your journal.
Cordially yours,

Hideki Kimura, M.D., Ph.D.
e-mail: hkimura@chiba-cc.jp